\begin{table}%t3
 \caption{Spin and shape models.}\label{Results}
%\centering
\par
\begin{tabular}{rcrrrrrrll}
\hline  \hline
Sidereal & Sense of & \multicolumn{2}{c}{Pole 1} & \multicolumn{2}{c}{Pole 2} & 
    $a/b$ & $b/c$ & Method & Reference \\
period (days) & rotation & $\lambda_{p}$ & $\beta_{p}$ & $\lambda_{p}$ & $\beta_{p}$ 
    &  &  & & \\
\hline
{\bf 52 Europa} & & & & & & & & & \\
   & & $0\degr$ & $37\degr$ & $203\degr$ & $38\degr$ & $1.12$ & $1.00$ & A & 
             Barucci et~al. (\cite{barucci86}) \\
   & & $17\degr$ & $65\degr$ & & & $1.11$ & $2.79$ & EAM & 
             Micha{\l}owski (\cite{michalowski93}) \\
$0.2346504$ & R & $80\degr$ & $-55\degr$ & $250\degr$ & $-40\degr$ & $1.21$ & $1.30$ & EA &
             Dotto et~al. (\cite{dotto95}) \\
$0.2347019$ & R & $84\degr$ & $-32\degr$ & $257\degr$ & $-18\degr$ & $1.20$ & $1.17$ & EAM &
             Micha{\l}owski et~al. (\cite{michalowski95}) \\
$0.2345855$ & P & $261\degr$ & $+60\degr$ & $63\degr$ & $+46\degr$ & $1.19$ & $2.2$ & EA &
             Lopez-Gonzalez \& Rodriguez (\cite{lopez99}) \\
$0.2346504$ & R & $79\degr$ & $-57\degr$ & $246\degr$ & $-44\degr$ & $1.2$ & $1.2$ & L &
             Kaasalainen et~al. (\cite{kaasalainen02})  \\
%\\
$0.2345813$ & P & $262\degr$ & $+46\degr$ & $71\degr$ & $+31\degr$ & $1.21$ & $1.04$ & EAM &
             Present work \\
$\pm 0.0000002$ & & $\pm 9\degr$ & $\pm 8\degr$ & $\pm 9\degr$ & $\pm 8\degr$ & $\pm 0.02$ &
    $\pm 0.05$ & & \\
$0.2345816$ & P & $252\degr$ & $+38\degr$ & $67\degr$ & $+25\degr$ & $1.15$ & $1.3$ & L &
            Present work \\
$\pm 0.0000002$ & & $\pm 5\degr$ & $\pm 5\degr$ & $\pm 5\degr$ & $\pm 5\degr$ & & & & \\	    
%\\
{\bf 115 Thyra} & & & & & & & & & \\
$0.301565$ & P & $175\degr$ & $+60\degr$ & $330\degr$ & $60\degr$ & $1.14$ & $1.30$ & EA &
             Dotto et~al. (\cite{dotto95}) \\
   &  & $197\degr$ & $30\degr$ & $358\degr$ & $35\degr$ & $1.22$ & $1.09$ & AM & 
             Blanco \& Riccioli (\cite{blanco98}) \\                 
%\\
$0.3017940$ &R& $182\degr$ & $-43\degr$ & & & $1.21$ & $1.03$ & EAM & Present work \\
$\pm 0.0000003$ & & $\pm 7\degr$ & $\pm 7\degr$ & & & $\pm 0.02$ & $\pm 0.03$ & & \\
$0.3017257$ &P& & & $7\degr$ & $34\degr$ & $1.23$ & $1.03$ & EAM & Present work \\
$\pm 0.0000003$ & & & & $\pm 7\degr$ & $\pm 7\degr$ & $\pm 0.02$ & $\pm 0.03$ & & \\
$0.3016652$ &P& & & $23\degr$ & $+33\degr$ & $1.1$ & $1.1$ & L & Present work \\
$\pm 0.0000002$ & & & & $\pm 8\degr$ & $\pm 4\degr$ & & & & \\
%\\
{\bf 382 Dodona} & & & & & & & & & \\
$0.17138450$ & P & $88\degr$ & $+68\degr$ & & & $1.54$ & $1.33$ & EAM & Present work \\
$\pm 0.00000007$ & & $\pm 4\degr$ & $\pm 4\degr$ & & & $\pm 0.01$ & $\pm 0.07$ & & \\
$0.17138442$ & P & $83\degr$ & $+64\degr$ & $248\degr$ & $+55\degr$ & $1.4$ & $1.3$ & L &
           Present work \\
$\pm 0.00000004$ & & $\pm 5\degr$ & $\pm 5\degr$ & $\pm 5\degr$ & $\pm 5\degr$ & & & & \\	    
\hline
\end{tabular}
\end{table}